Author's response to reviews

Title: The role of entry screening in case finding of tuberculosis among asylum seekers in Norway

Authors:

Ingunn Harstad (ingunn.harstad@ntnu.no)
Geir W Jacobsen (geir.jacobsen@ntnu.no)
Einar Heldal (einar.heldal@c2i.net)
Brita A Winje (brita.askeland.winje@fhi.no)
Saeed Vahedi (saeed.vahedi@ulleval.no)
Anne-Sofie Helvik (Anne-Sofie.Helvik@ntnu.no)
Sigurd L Steinshamn (sigurd.steinshamn@ntnu.no)
Helge Garåsen (helge.garasen@ntnu.no)

Version: 4 Date: 4 November 2010

Author's response to reviews:

About formatting changes: The major revisions have been done, some of the minor revisions have been done but the tables have not been changed, and I am not sure if the reference 20 is correct.